Notes of the RICH1 Phone meeting 29/09/2006
People Present:
CERN:  Kenneth, Olav, Werner
IC:         Dave, Gabriel, Richard, Bill
RAL:      Peter

Bristol:   Richard
Apologies for absence:     Mike, Derek, Trevor, Geoff, Neville, Fabio
Discussed Agenda Topics:
1. Gas Enclosure

There are still several small items to be completed on the GE including locking its position on the mounts (temporary stops are installed at present), designing and fitting a restraint to prevent tipping and replacing a damaged door O-ring. Tony has confirmed that he will supply the O-ring material. The Imperial team would also like to survey the position of the quartz window apertures to allow correct positioning of the HPD arrays, which are mounted on the iron shields.

2. Exit Window
The hole has now been cut in the EW and the boss will be shipped to RAL by the manufacturer on Monday. The plan is to send both directly to CERN where Derek, who is there for other reasons, will do the gluing during the week starting October 9th. Rolf has confirmed that the installation is foreseen during the week starting October 16th, with the 17th and 18th being most likely. The installation tooling is already at CERN, stored in B156, and draft PSSPS documents are available on EDMS.

On the diaphragm Peter reported that they have been testing a new pigment from GE and that he had posted two samples to Dave on Tuesday. They are the new GE pigment in the Dow-Corning base material and the same pigment in the GE base material. The standard mix is 2% so they have tried 5% for these samples. It looks very black and the mechanical properties seem good. (After the meeting the samples arrived with Dave, who has now processed them and can measure no transmission, implying an attenuation of at least 5x10**7.) Testing at bake-out temperatures has still to be done. The previous tests with the Dow-Corning base material were successful but the diaphragm could not subsequently be removed from the dummy beam-pipe. Werner was anxious that time was short and it was agreed not to wait for the diaphragm before shipping the exit window. Peter confirmed that it takes only 24 hours to produce one unit.
3. Beampipe
Werner reported that the beam-pipe was installed and testing was in progress. There was still some concern over a “virtual” leak. Dave asked for confirmation of the position of the exit window connection fin but since the protection is still in place this can not be provided. However with the new diaphragm material it should be less critical. The fixtures for the support wires have still to be installed.

4. Quartz Windows
Dave reported that the 4 panes had been shipped to CERN and were due to arrive today. They are out of tolerance by about 1.2mm in length but since Trevor has confirmed that this can be accommodated, they were accepted anyway, to avoid further risk of breakage. The Imperial team is considering refinements to the installation schedule to minimize the risk to both the windows and the beam-pipe. The upper window could be installed this year, with the beam-pipe protection still in place, while the lower one would be pushed later into next year. He understands that the coating process is scheduled for November.

5. Spherical Mirrors

Kenneth reported that the optical properties (D0, radius and reflectivity) of the samples which have now received a ten-year radiation dose are unchanged from the starting measurements. The mechanical tests have not yet been done. The sample in the C4F10 tank will be tested again in about 2 weeks after completing 4 months exposure.
The CMA samples coated at CERN produced very good reflectivity results, comparable with a glass sample also coated. Dave requested that André should try to remove the coating from one sample and re-do it to verify the technique. CMA are not yet ready to try coating themselves. The tank has been modified but a necessary technician will not be available until next week.

The carbon-fibre material is available at CMA and some of the other pieces are in production. The mandrel polishing is scheduled to be completed rather soon. Kenneth will try to arrange a telephone meeting for Wednesday October 4th although he himself will be on vacation.
The possible option of coating the mirrors at CERN has been taken no further for the moment.

6. Flat Mirrors
Some pieces required for the mounting and alignment are still awaiting manufacture at Imperial. Richard said that he would give priority to the alignment rig for the moment but that he would ensure that it was compatible with his ideas for the installation. Ideally a meeting with Richard, Nick and the Imperial people should be arranged when Nick is again available.
7. HPD Assembly
Gabriel remarked that in the four weeks since the last meeting significant progress had been made. The material for the ladder rungs arrived and their production has started in the Imperial workshop. The U-channel pieces are modeled and drawings are in preparation. Four prototypes, one of each design, (2 ladders) will be manufactured. The extraction mechanism has been designed and the cooling plate dimensions fixed. The copper for the cooling plates is in stock but the honeycomb must be ordered soon and this depends on the manifold design which has still to be finalized. The square copper tube has arrived in CERN and the connection pieces produced. Gabriel has still to confirm the availability of the Genoa technicians with Mario but is expecting to start production in mid-November. Olav requested that Erich be kept well-informed of the plans for the production. Dave noted that the cooling plates were not required to dress and test a column and so prototype columns, that need to be tested before the PPR is held, could be dispatched to CERN without the cooling plates. 
The magnetic shields and kapton insulation are available in CERN. Two Imperial technicians, Vera Kasey and Maria Khaleeq, will go to CERN with Richard Plackett for two weeks, starting on October 16th to be trained by Tito and to wrap as many shields as possible in the time available. Lab space for this job has been reserved by Tito.

The patch panel design is close to final and Gabriel has fixed the length of the fibre ribbon break-outs so that these can be ordered by Ken. However the external cable routing and the possible use of cable chains is largely undefined. Gabriel will try to arrange preliminary discussions with Rolf and Olivier (morning of October 16th).
Rohan came to Imperial in early August to discuss his latest ideas for the HPD installation and will need to be kept informed as the design becomes firmer.
8.  Magnetic Distortions

Ray has not yet responded to Trevor’s proposed space envelope. A meeting should be arranged to review the current status. Olav reminded the meeting that the DCS interface to this device would have to be defined, and that Flavio should be invited to the meeting. 
9.  C4F10

No news from Oxford.
10. Aerogel

Olav expressed concern at some information reported in Heidelberg. After reconstruction the average number of aerogel photons, according to Chris Jones, is 3.8. Dave said that there were still inconsistencies to be resolved. Clara also reported a factor of two decrease in clarity after immersion of the aerogel in C4F10. Dave proposed that this topic should be discussed at the next general RICH meeting.
11. Next meeting
Friday 13th October 2006 at 10:00(GMT) 11:00(CET). 
